Accelerator Status Report

Run Schedule: Jan. 3, 2011 through May 13, 2011
March 23, 2011
Injector    Energy 61.76 MeV
· (3/22)  Beam Studies: checked and corrected the x-y coupling in the injector.

· (3/19) The 45MeV spectrometer mismatched. EES fixed the power supply. 
· (3/18) The Capture gradient was not stable causing beam to trip. EES replaced a 2.5W amplifier. 
· (3/11) The injector optics was optimized. The injector was set up for high current run (150uA). 
Accelerator   Linac Energy 549.00 MeV
· (3/21) VBV2L16A would not open. The trip level was adjusted to open the valve.
· (3/19) The ARC2A power supply started to be unstable. After many hours of troubleshooting it was found that the connection from the power supply to the magnets at the power supply end was loose. The loose connection made the power supply unstable, which generated a lot heat inside the power supply, and the power supply was severely damaged due to overheating. About 46 hours was lost. While troubleshooting the ARC2A some beam studies and maintenance were conducted. 
· The accelerator did not deliver much beam last week due to the activities in the halls. The time was utilized for maintenance and improving the machine optics. Over the weekend we delivered low current to Halls A and C. 
· (3/15) Beam delivery was suspended for about 12 minutes to conduct a tornado drill. The drill went well. After the drill, four hours beam studies: Commissioning Helicity Magnets at 1kHz Helicity Reversal Rate. 
· (3/15) Four cavities in 1L20 were recovered.  
· (3/12) Zone 1L20 was bypassed. 

Hall A---Pass 4, E08 - 010
· The beam was fine.
· (3/15) Q1 is fixed.

· The hall did not take beam until Sunday owl shift (3/13). The hall had problems with its left Q1 magnet. The magnet would not ramp up to the desired current. 
Hall B ---Pass 5, TPE
· The hall has been down installing a new experiment (HDice).
Hall C--- Pass 1, QWeak 
· (3/19) The vacuum in the dump area started to deteriorate and the hall has been down since. The problem is not directly beam related. The hall personnel are still investigating the problem. The radiation level has been high, which makes the investigation difficult.
· The hall reported that their beam was good even their halo counts were higher than the limits.
· The hall started to take beam Friday swing shift (3/11). It has been taking low current (< 25uA). The beam was fine. The BLA was calibrated at 70uA and needs to be calibrated at higher current (150uA). The hall will be down today to cool down its target.
· (3/15) MQE3M03 power supply would not turn on. Hall A is running with both Moeller quads off. EES is helping troubleshoot the problem. Hall C beam size will be large until the Moller quads are fixed.
